Argonne National Laboratory9700 Cass AvenueArgonne, Illinois 60439DOCUMENT NO. 410201-00092-01TECHNICAL SPECIFICATIONFORFRONT END SUPPORT TABLES, STAGESAND KINEMATIC MOUNTSWBS 1.4.1.2.2.1.7WBS 1.4.1.2.2.1.8WBS 1.4.1.2.1.1.10WBS 1.4.1.2.1.1.11

A R G O N N E N A T I O N A L L A B O R A T O R Y 410201-00092

Title: Technical Specification for Front End Support Tables,Stages and Kinematic MountsR e v . 01ApprovedDate

Page

6˚ of

˚ ˚ 6 ˚ ˚

ANL-242 (3-83)4.2.6

Acceptance Testing. The acceptance testing shall be performed inaccordance with the ANL approved contractor’s test plan, and the results submitted tothe Laboratory for approval. The following items are the minimum requirements forinclusion in the acceptance tests.4.2.6.1 Verification of Fiducial Holes. The size and location of fiducialholes for each assembly will be verified for compliance with the engineering drawings.4.2.6.2 Verification of Overall Dimensions. The maximum width andheight of the assemblies with respect to the appropriate reference surfaces shall beverified to be in conformance with the assembly drawing for the item.4.2.6.3 Verification of Axis motion. The motion of each moving oradjustable axis shall be tested for the following: Range of motion, Binding, Locking,and where applicable Trueness of motion (square, straight, and linear), andRepeatability.4.2.6.4 Verification of Load Handling Capability. The ability of thedevice to handle its rated load shall be tested and verified for static and dynamicconditions. Axis motion (section 4.2.6.3 above) must be verified under rated load.Testing of axis motion without load is optional.5.0

PREPARATION FOR DELIVERY5.1 Each assembly shipped shall be identified by the appropriate purchase ordernumber, assembly number and a sequential identity or serial number.5 . 2 Any moving components on an assembly must be firmly secured for shipmentin a manner that precludes damage from motion and vibration.5 . 3 Assemblies must be secured for shipment in fashion that protects the assemblyfrom dust, dirt and weather or water damage. The assemblies must be securelyfastened, padded, boxed and/or crated for shipment in a suitable manner for preventionof shipment damage to any assembly and its weather/dust protection and insure theirsafe delivery to the Laboratory..